Author's response to reviews

Title: Association between obesity and metabolic co-morbidities among children and adolescents in South Korea: the 2007-2008 Korea National Health and Nutrition Examination Surveys

Authors:

HYUNJUNG LIM (hlim@jhsph.edu)
HONG XUE (hxue@jhsph.edu)
YOUFA WANG (ywang@jhsph.edu)

Version: 2 Date: 29 May 2013

Author's response to reviews:

We provided the information about ethics committee approval for this study. The study is based on existing survey data in public domain in South Korea.